Author's response to reviews

Title: Right Lower Lobectomy Eight Years After Left Pneumonectomy for a Second Primary Lung Cance

Authors:

Yunpeng Liu (rhame-snow@163.com)
Peipeng Cui (cpp1778@163.com)
Zhiguang Yang (13504303387@139.com)
Peng Zhang (40619693@qq.com)
Rui Guo (28277832@qq.com)
Guoguang Shao (guoguangshao@sohu.com)

Version: 4 Date: 8 March 2013

Author's response to reviews: see over
March 7th, 2013

To the Editor of the Journal of Cardiothoracic Surgery

RE : Right Lower Lobectomy Eight Years after Left Pneumonectomy for a Second Primary Lung Cancer. Yunpeng Liu et al. MS : 1591300354859246

To the Editor :

We would like to resubmit this paper for possible publication in the Journal of cardiothoracic Surgery. Based on the reviewer’s recommendations, we have :

1. Rewritten the entire manuscript in better English.

2. We have addressed the issue of the vascular (pulmonary hypertension) aspect of lung function.

3. We looked again at the literature and discussed it differently taking into account whether surgery was done for benign or malignant disease.

4. We have addressed the issue of N2 disease.

5. In the text and in our conclusion, we stress that surgeons should be cautious and careful before doing such cases.

We would also like to thank the reviewers for their useful and thoughtful comments. Because of those, the current manuscript is much improved over the previous one.

Guoguang Shao, m.d.
Corresponding Author